Dear Editor:

Thank you very much for giving us an opportunity to revise our manuscript. We appreciate the editor and reviewers very much for their constructive comments and suggestions on our manuscript entitled “Mantle Cell Lymphoma with Endobronchial Involvement: A Case Report” (ID: 72068).

We have studied the comments carefully and made corrections which we hope meet with approval. The main corrections are in the manuscript and the responds to the reviewers’ comments are as follows.

Reviewer #1: A nice case presentation was made that will be useful to clinicians.

Response: Thank you very much for your affirmation.

Reviewer #2: Overall, the manuscript was an interesting read and relevant to its field. The manuscript is well organized and follows a clear flow. Only language and grammar editing are necessary.

Response: We have sent our manuscript to AJE as you recommend and the language editing certificate was uploaded.

Science editor: This report presents a rare case of mantle cell lymphoma (MCL) with endobronchial involvement. This case report is novel, but the
writing is not standardized enough. Please provide detailed case data and discuss the feasibility of treatment strategies for mantle cell lymphoma with endobronchial involvement in the discussion section.

Response: We discussed the feasibility of treatment strategies for mantle cell lymphoma with endobronchial involvement in the discussion section in our revision.

Company editor-in-chief: Please provide the original figure documents. Please prepare and arrange the figures using PowerPoint to ensure that all graphs or arrows or text portions can be reprocessed by the editor.

Response: We have uploaded our original figure documents.

Once again, thank you very much for your constructive comments and suggestions which would help us both in English and in depth to improve the quality of the paper.

Kind regards,

Zhao Xiaojing

E-mail: 18133@renji.com